Author's response to reviews

Title: New technical approach for Abdominal wall defects after TRAM-flap: a case report

Authors:

Daniel A Kaemmer (dkaemmer@ukaachen.de)
Joachim Conze (jconze@ukaachen.de)
Jens Otto (jeotto@ukaachen.de)
Volker Schumpelick (vschumpelick@ukaachen.de)

Version: 5 Date: 18 February 2008

Author's response to reviews: see over
Dear Editor-in-Chief,

I regret to answer you that late, but due to holidays I was out of office for some time.

I submitted a revised version of the manuscript as you proposed, regarding the title page and patients’ consent section. Again all changes made have been marked in yellow.

Regarding the number of references I again checked all of them and would be very pleased if you would agree to add one more reference than normally allowed. Because of the short review on different techniques of breast reconstructing and already known complications as well as an outlook on interesting alternatives as IPOM technique proposed by Mr Booi and new mesh materials I do not think that any reference is redundant.

Please do not hesitate to contact me in case of any further questions.

Sincerely,

Dr. med. D.A. Kaemmer